Author's response to reviews

Title: Adult vaccination coverage levels among users of complementary/alternative medicine--results from the 2002 National Health Interview Survey (NHIS)

Authors:

Shannon Stokley (sstokley@cdc.gov)
Karen A Cullen (kcullen@cdc.gov)
Allison Kennedy (akennedy@cdc.gov)
Barbara H Bardenheier (bbardenheier@cdc.gov)

Version: 3 Date: 30 January 2008

Author's response to reviews: see over
January 30, 2008

RE: MS 1341980068154919 Adult Vaccination Coverage Levels Among Users of Complementary/Alternative Medicine--Results from the 2002 National Health Interview Survey (NHIS)

Dear Editor, BMC Complementary and Alternative Medicine,

We are pleased to hear that our manuscript has been accepted for publication. Based on your suggestions, we made the following formatting changes:

- Changed the title of the manuscript to sentence case
- Included the country in the affiliation details
- Listed all author names for references 3 and 29
- Removed the second set of column headings in Table 2

Upon the final review of the manuscript we noticed that Table 4 had been deleted from the last version submitted to BMC. We have added the table back to the manuscript.

If there are other corrections that need to be made, please feel free to contact me. We look forward to seeing the published manuscript!

Thank you,

Shannon Stokley, MPH
Epidemiologist
National Center for Immunization and Respiratory Diseases (NCIRD)
Centers for Disease Control and Prevention (CDC)
Routine Mail: 1600 Clifton Road NE, MS E-52, Atlanta, Georgia 30333
Fed Ex Mail: 12 Corporate Square Blvd, Room 4301, Atlanta, GA 30329
Telephone: 404/639-8734
FAX: 404/639-8614
Email: sstokley@cdc.gov